Author's response to reviews

Title: Microbiology of airway disease in a cohort of patients with Cystic Fibrosis

Authors:

Antonietta Lambiase (alambias@unina.it)
Valeria Raia (raia@unina.it)
Mariassunta Del Pezzo (delpezzo@unina.it)
Alfredo Lavitola (lavitola@unina.it)
Angela Sepe (ornellasepe@hotmail.com)
Fabio Rossano (fabio.rossano@unina.it)

Version: 2 Date: 28 July 2005

Author's response to reviews:

To the Editor of BMC Infectious Diseases

We thank you for your consideration of our paper entitled "Microbiology of airway disease in a cohort of patients with Cystic Fibrosis" previously submitted to "BMC Infectious Diseases". We modified all the text according to the reviewers queries and we revised all the manuscript point by point.

In the revised version of the paper, the contribution of Dr. Vincenzo Carnovale has been particularly relevant for inclusion of the other clinical data, therefore we ask to include him as a coauthor of the paper.

Sincerely,

Dr. Antonietta Lambiase